Author's response to reviews

Title: Mutation analysis of genes that control the G1/S cell cycle in melanoma: TP53, CDKN1A, CDKN2A, and CDKN2B

Authors:

Jose Luis Soto (miguel.lopez.nevot.sspa@juntadeandalucia.es)
Carmen M. Cabrera (mcabrm@fundacionhvn.org)
Sabio Serrano (miguel.lopez.nevot.sspa@juntadeandalucia.es)
Miguel Angel Lopez-Nevot (miguel.lopez.nevot.sspa@juntadeandalucia.es)

Version: 4 Date: 6 April 2005

Author's response to reviews:

Formatting changes requested:

1. The abstract has been amended.
2. The highlighting and tracking have been removed from the text.
3. The figures have been cropped.